Author's response to reviews

Title: Short term micronutrient-antioxidant supplementation has no impact on a serological marker of gastric atrophy in Zambian adults: retrospective analysis of a randomised controlled trial

Authors:

Violet Kayamba (viojole@yahoo.com)
Mumba Chomba (mumbaschomba@gmail.com)
Paul Kelly (m.p.kelly@qmul.ac.uk)

Version: 5 Date: 13 March 2014

Author's response to reviews:

Tropical Gastroenterology
& Nutrition group
(TROPGAN)
Department of Medicine,
University of Zambia School of Medicine,
UTH, Nationalist Road,
PO Box 50398
Lusaka, Zambia
Dr Violet Jolezya Kayamba BScHB MBChB MMed
viojole@yahoo.com

The Editor,
BMC Gastroenterology

Dear Sir/Madam,

Manuscript revision

Thank you for considering our manuscript Short term micronutrient-antioxidant supplementation has no impact on a serological marker of gastric atrophy in
Zambian adults: retrospective analysis of a randomised controlled trial for publication in BMC gastroenterology.

We have thoroughly gone through the manuscript and corrected areas that had some errors. We have also edited the references to suit the exact format required. Professor Paul Kelly who is the senior author has written a separate letter outlining his views on the manuscript.

We greatly appreciate your response and hope to hear from you soon.

Yours Sincerely,

Violet Jolezya Kayamba.